Author’s response to reviews

Title: Emergence of New Biovars and Sequence Types of Brucella isolates, Inner Mongolia, China

Authors:

Yanfen Chen (chenyanfen1015@163.com)
Yuehua Ke (yuehuake@yahoo.com.cn)
Yufei Wang (yufeiwang21@yahoo.com)
Xitong Yuan (Y100039@yahoo.com.cn)
Xiaoyan Zhou (leiping2222@163.com)
Hai Jiang (jianghai@icdc.cn)
Zhoujia Wang (grce-zhuji@163.com)
Qing Zhen (zq415@sina.com)
Yaqin Yu (yuyaqin5540@163.com)
Liuyu Huang (huangly@nic.bmi.ac.cn)
Buyun Cui (cuibuyun@icdc.cn)
Zeliang Chen (zeliangchen@yahoo.com)

Version: 3 Date: 15 August 2012

Author’s response to reviews: see over
Dear Editor,

We have revised the manuscript according to the style requirement of journal. The revision include the following:
1. Abstract was revised into structured format, background, methods, results, conclusions are included.
2. Competing interests section was added.
3. Author contributions section was added.

Sincerely yours

Zeliang Chen